Author’s response to reviews

Title: Can patients make heads or tails of Enhanced Primary Health Care (EnPHC)? Experience through their own journey

Authors:

Mohammad Zabri Johari (mzabrijohari@gmail.com)

Zalilah Abdullah (drzalilah@gmail.com)

Ainul Nadziha Mohd Hanafiah (ainulnadziha.mh@moh.gov.my)

Nur Izzati Mohammed Nadzri (izzatinadzri91@gmail.com)

Siti Aisyah Razli (aisyah.razli93@gmail.com)

Yuke Lin Kong (kong.yl@moh.gov.my)

Version: 4 Date: 10 Aug 2020

Author’s response to reviews:

Dear Editor and Reviewers,

We thank you for the comments and suggestions from the previous review. We thank you the Reviewers for their constructive comments and their acceptance of revision made. For this revision, we have made changes based on the Editors comments:

1. All text requires careful editing for grammar, wording and typos.
   - We have went through the manuscript carefully and made changes to correct errors and comply with UK English.
   - We also made updates on acronyms used in the manuscript.

2. Please, specify type of the qualitative study.
   - The study is a phenomenology qualitative study that is mentioned now in both the abstract (line 35) and the study design (line 115).

3. Please, explain rigor and trustworthiness of the research.
   - Rigor and trustworthiness is explained as a new section in the methodology (line 166-180) and is following criteria by Lincoln and Guba.

4. Please, consider citing references uniformly.
   - The citations has been updated uniformly using EndNote X9.

5. All changes should be coloured red.
- All changes made is marked in red to ease revision.

We hope that all changes are acceptable for the purpose of publication. We hope to hear the good news soon.

Best wishes,

Mohammad Zabri Johari, Primary Author on behalf of the team.